Dear Editors and Reviewers:

Thank you for your letter and for the reviewers’ comments concerning our manuscript entitled “Effects of ulinastatin therapy in deep vein thrombosis prevention after brain tumor surgery: a single-center randomized controlled trial” (MS No: 87702). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made a correction which we hope meets with approval.

We have addressed the comments raised by the reviewers, and the amendments are highlighted in yellow in the revised manuscript, Title page, abstract et. Point by point responses to the reviewers’ comments is listed below this letter. Thank you! We hope that the revised version of the manuscript is now acceptable for publication in your journal.

I look forward to hearing from you soon.

With best wishes,

Yours sincerely,

Yuhai Wang

We would like to express our sincere thanks to the reviewers for their constructive and positive comments.

Replies to Reviewers

Specific Comments

Responds to the reviewer’s comments:

Reviewer #1:

Scientific Quality: Grade C (Good)

Language Quality: Grade B (Minor language polishing)

Conclusion: Minor revision

Specific Comments to Authors: Dear authors the presented article is interesting. It is necessary to make a comparison with LMWH which can be monitored and dose adjusted. Best regards

Author Response: Thank you very much for your review and high evaluation. Your
comments were very important and meaningful. Firstly, we do not routinely use LMWH to prevent DVT, it may increase the risk of bleeding after brain tumor surgery; secondly, the present study was a single-center randomized controlled trial, the study protocol and Outcome evaluation were prepared in advance. However, we will design a new single-center randomized controlled trial to study LMWH, and compare the effective and safety of LMWH and UTI. Thank you very much again! Additionally, we had sent our revised manuscript to a professional English language editing company to polish the manuscript further. Thank you very much for your review and comments.

Thank you very much for your comments again.

Yuhai Wang
9-Oct-23